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Do Data Taking

Day Delivered  Recorded  Efficiency Comment
Lum (pb™)  Lum (pb?) (%)

30 Jul 6.27 5.82 93

31 Jul 6.81 6.27 92

01 Aug 4.36 4,06 93

02 Aug 7.08 6.07 86 High humidity leads to tripped HV.
03 Aug 4.09 3.66 90

1 hour down for datalogger disk errors.
04 Aug 4,05 2.77 68 Additional downtimes for special run and HV
trips

05 Aug 0.00 0.00 Cosmic runs taken.
30 Jul -

S Aug 32.7 28.6 88

July 142.0 129.0 01 Record month for efficiency

3" best for delivered and recorded Luminosity
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Luminosity (/fb)
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192 April 2002 - 5 August 2007
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3.29

efficiency = 87%

Since the last shutdown:
Delivered Lumi = 1.68 tb-1
Recorded Lumi = 1.46 tb-1
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Efficiency

Daily Data Taking Efficiency!

FYO07 through 5 August 2007
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Shutdown News

* On Sunday at 19:00, DO went into supervised access
— All systems were shutdown by 21:00 for the power outage.

* After the power outage this morning

— The DAQ system was up by 12:00.

— Half of the systems have been brought up without any
problems at this time.

® Current plans

— For the next two days the alignment group will be surveying
the hall.

— On Thursday we will open the detector.
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